High Prevalence of Skin Disorders among HTLV-1 Infected Individuals Independent of Clinical Status

Renata Okajima^{1,2}, Augusto C. P. Oliveira¹, Jerusa Smid¹, Jorge Casseb^{1,3¶}*, Jose Antonio Sanches, Jr.^{2¶}

1 HTLV-Outpatient Clinic, Institute of Infectious Diseases "Emilio Ribas," São Paulo, Brazil, 2 Department of Dermatology, University of São Paulo Medical School, São Paulo, Brazil, 3 Institute of Tropical Medicine of São Paulo, Laboratory of Medical Investigation 56 (LIM56), University of São Paulo Medical School, São Paulo, Brazil

Abstract

Background: Human T-cell lymphotropic virus type 1 (HTLV-1) infection can increase the risk of developing skin disorders. This study evaluated the correlation between HTLV-1 proviral load and CD4⁺ and CD8⁺ T cells count among HTLV-1 infected individuals, with or without skin disorders (SD) associated with HTLV-1 infection [SD-HTLV-1: xerosis/ichthyosis, seborrheic dermatitis or infective dermatitis associated to HTLV-1 (IDH)].

Methods: A total of 193 HTLV-1-infected subjects underwent an interview, dermatological examination, initial HTLV-1 proviral load assay, CD4⁺ and CD8⁺ T cells count, and lymphproliferation assay (LPA).

Results: A total of 147 patients had an abnormal skin condition; 116 (79%) of them also had SD-HTLV-1 and 21% had other dermatological diagnoses. The most prevalent SD-HTLV-1 was xerosis/acquired ichthyosis (48%), followed by seborrheic dermatitis (28%). Patients with SD-HTLV-1 were older (51 vs. 47 years), had a higher prevalence of myelopathy/tropical spastic paraparesis (HAM/TSP) (75%), and had an increased first HTLV-1 proviral load and basal LPA compared with patients without SD-HTLV-1. When excluding HAM/TSP patients, the first HTLV-1 proviral load of SD-HTLV-1 individuals remains higher than no SD-HTLV-1 patients.

Conclusions: There was a high prevalence of skin disorders (76%) among HTLV-1-infected individuals, regardless of clinical status, and 60% of these diseases are considered skin disease associated with HTLV-1 infection.

Citation: Okajima R, Oliveira ACP, Smid J, Casseb J, Sanches JA Jr (2013) High Prevalence of Skin Disorders among HTLV-1 Infected Individuals Independent of Clinical Status. PLoS Negl Trop Dis 7(11): e2546. doi:10.1371/journal.pntd.0002546

Editor: Joseph Raymond Zunt, University of Washington, United States of America

Received May 29, 2013; Accepted October 4, 2013; Published November 7, 2013

Copyright: © 2013 Okajima et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This work was partly supported by Fundação de Amparo à Pesquisa o Estado de São Paulo (FAPESP), Grant 2008/56427-4. JC is senior researcher from the National Research Council of Brazil. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: jcasseb10@gmail.com

¶ Both JC and JAS were principal investigators of this study.

Introduction

Adult T-cell leukemia/lymphoma (ATLL), HTLV-1-associated myelopathy/tropical spastic paraparesis (HAM/TSP) and infective dermatitis associated with HTLV-1 (IDH) are the main diseases caused by human T-cell lymphotropic virus type 1 (HTLV-1) infection [1–3]. However, several other clinical conditions have been associated with this viral infection, such as uveitis, thyroiditis, arthritis and polymyositis [4–6].

There are an estimated 5 to 10 million HTLV-1 infected individuals worldwide and Brazil is considered a highly endemic area for HTLV-1 infection, with the largest absolute number of HTLV-1 infected individuals, with more than one million people living with this virus [7–9]. Despite this high prevalence, only a few studies on the dermatological aspects of HTLV-1 infection have been described in this country [10].

There is a lack of surrogate markers to assess the infected patients who have a higher risk for HTLV-1 associated skin disorders. Moreover, there are few immunological studies among HTLV-1-infected persons who are simultaneously suffering from HAM/TSP and skin diseases. The aim of this study is to evaluate the prevalence of skin disorders in HTLV-1infected individuals and to correlate this prevalence with the initial HTLV-1 proviral load, and initial $CD4^+$ and $CD8^+T$ cell count.

Materials and Methods

In the last 18 years, a cohort of HTLV-infected subjects has been followed in the HTLV-outpatient clinic at the Institute of Infectious Diseases "Emilio Ribas" (IIER), with the support of nurses, nutritionists and physical therapists. From a total 450 HTLV-1-infected individuals, including asymptomatic carriers and HAM/TSP patients, 193 of them were consecutively evaluated for skin disorders from January 2008 to July 2010 by the same dermatologist, blinded for the clinical status to minimize information bias. Demographical and clinical dates were collected, and dermatological examinations were carried out. The HIV coinfected individuals were excluded, but HCV co-infected subjects were included.

Ethics Statement: Written informed consent was obtained from all participants, and the IIER ethical board approved the protocol.

The patients underwent laboratory studies, including HTLV-1 serological diagnosis, initial CD4⁺ and CD8⁺T cell counts, and an

Author Summary

HTLV-1 infection may increase the risk of developing skin disorders. A total of 193 HTLV-1 infected subjects were studied, including asymptomatic carriers and HAM/TSP patients. Of the subjects, 76% had an abnormal skin condition, with a high prevalence both among HTLV-1 asymptomatic carriers and HAM/TSP patients. The most prevalent SD-HTLV-1 was xerosis/acquired ichthyosis (48%), followed by seborrheic dermatitis (28%). Patients with SD-HTLV-1 were older (51 vs. 47 years), had a higher prevalence of myelopathy/tropical spastic paraparesis (HAM/TSP) (75%) and an increased first HTLV-1 proviral load compared with patients without SD-HTLV-1. When excluding HAM/TSP patients, the first HTLV-1 proviral load of SD-HTLV-1 individuals remains higher than no SD-HTLV-1 patients. Thus, skin diseases are highly prevalent among HTLV-1-infected individuals.

initial HTLV-1 proviral load. In accordance with previous studies, the following skin disorders associated with HTLV-1 infection (SD-HTLV-1) were considered: xerosis/acquired ichthyosis, seborrheic dermatitis and infective dermatitis associated with HTLV-1 (IDH) [1,10-12]. HAM/TSP was diagnosed according to previously established criteria [13]. Skin culture and punch skin biopsies were performed when clinical examination was not sufficient for a dermatological diagnosis. Antibodies to HTLV-1/2 were detected by a diagnostic enzyme-linked immunosorbent assay (ELISA) and confirmed by Western blot analysis and polymerase chain reaction (PCR), which are capable of discriminating between HTLV-1 and HTLV-2 [14]. To determine the counts of CD4⁺ and CD8⁺ T-cell subsets, fresh whole blood specimens were collected in EDTA tubes and subjected to flow cytometry (Coulter EPICS® XL-MCLÔ Flow Cytometer -Beckman Coulter, Fullerton, CA), using human monoclonal antibodies anti-CD3, anti-CD4, and anti-CD8, labeled with fluorochrome.

The results of the first HTLV-1 proviral load were available in the database. Quantitative proviral DNA levels were detected by a real-time automated PCR method, using TaqMan probes for the *pol* gene. The albumin gene served as the internal genomic control, and MT2 cells were used as a positive control. The results are reported as copies/10000 PBMCs, and the detection limit was 10 copies [15].

Data were analyzed using SPSS 17.0 software. The association between independent variables and the outcome was analyzed either by Student's t-test or ANOVA (normal distribution variables), or by Mann-Whitney test (non-normal distribution variables), while the association between categorical variables and the outcome was assessed by the X² test. HTLV-1 proviral load was log-transformed to obtain a normal distribution. Correlations between HTLV-1 proviral load and SD-HTLV were performed using Spearman's rank correlation. Data are expressed as mean \pm standard deviation (normal distribution variables) or median and interquartile range (non-normal distribution variables). Statistical significance was set at a p value<0.05.

Results

One hundred ninety-three HTLV-1-infected subjects (43% of all HTLV-1-infected patients at the HTLV outpatient clinic from the Emilio Ribas Institute cohort) underwent a dermatological exam. Mean age of patients was 49.4 ± 12.3 years. Female gender had a higher prevalence of HTLV-1 infection (72%).

Regarding the presence of neurological involvement, 38% of the patients had a diagnosis of HAM/TSP. The dermatological examination revealed a high prevalence of skin disorders among the HTLV-1-infected patients (76%). Sixty-five individuals (34%) had one dermatological condition, and 42% (n=81) of the patients had two or more dermatological conditions. Among the 147 patients that had an abnormal skin condition, 79% (n=116) had a skin disorder associated with HTLV-1 infection (SD-HTLV-1) (xerosis/ichthyosis or seborrheic dermatitis) and 21% (n=31) had other dermatological diagnoses. The most prevalent skin disorder associated with an HTLV-1 diagnosis was xerosis/ichthyosis (48%), followed by seborrheic dermatitis (28%).

Table 1 shows the prevalence of skin disorders in the HTLV-1 patients based on a diagnosis of HAM/TSP or asymptomatic carriers. SD-HTLV-1 were more prevalent on HAM/TSP patients (xerosis/acquired ichthyosis (p = 0.007; seborrheic dermatitis (p = <0.0001); IDH (p = 0.022). Patients with SD-HTLV-1, including asymptomatic carriers and HAM/TSP, are older, have a higher prevalence of HAM/TSP, and have a higher first HTLV-1 proviral load (performed for 109 patients, p = 0.009), compared with patients without SD-HTLV-1 (Table 2). Note that 75% of the SD-HTLV-1 group was made up of HAM/TSP individuals.

Table 3 depicts the presence of SD-HTLV-1 in subjects that are asymptomatic for neurological symptoms (absence of HAM/TSP). Mean age of SD-HTLV-1 patients was 51 years, as compared with 44 years for HTLV-1 patients without SD (p = 0.002), regardless of gender, CD4⁺ and CD8⁺ T cell counts (p = 0.489; p = 0.824). The initial HTLV-1 proviral load was significantly higher for the group with SD-HTLV-1 as compared with that for the group without SD-HTLV-1 and asymptomatic for neurological symptoms (p = 0.021).

Discussion

Notably, 76% of the HTLV-1-infected asymptomatic carriers and 88% of the HAM/TSP patients showed some skin disorder in our study. These findings are similar to those described in two previous studies involving asymptomatic carriers and HAM/TSP subjects [10,12]. Thus, it is important to stress that HTLV-1 infection may have an etiological link to skin disease. In fact, skin disorders are highly associated with HTLV-1 infection, regardless of neurological symptoms, and they may represent a clinical warning sign for the diagnosis or progression of this infection [10– 12].

Excluding HAM/TSP cases, subjects with a diagnosis of SD-HTLV-1 are older than groups with other types of skin disorders or individuals with a normal dermatological exam. However, no significant association was observed with gender. These findings suggest that older HTLV-1-infected individuals, who probably have a longer duration of their viral infection have had more time to develop SD-HTLV-1.

Although IDH is the only skin disease in which HTLV-1 infection is a criterion for diagnosis, other skin disorders could also be associated with HTLV-1 infection, including xerosis/acquired ichthyosis and seborrheic dermatitis, as previously demonstrated [10–12,16]. In fact, in our study, these illnesses were the most prevalent skin disorders associated with adult HTLV-1-infected individuals, regardless of the clinical status. However, studies with a longer follow-up should be performed to assess the hypothesis that these skin manifestations are related to HTLV-1 infection.

The exact pathogenic mechanism of IDH still needs to be made clear, but the current view is that a diagnosis of HTLV-1 infection

Table 1. Dermatological findings among human T-cell lymphotropic virus type-1 (HTLV-1) infected individuals (HAM/TSP and asymptomatic carriers).

Dermatological findings	HAM/TSP N (%)	HTLV-1 Asymptomatic carriers N (%)	p value
	n = 73	n = 120	
Xerosis/Acquired ichthyosis	44 (60.2)	51 (42.5)	0.006
Seborrheic dermatitis	35 (47.9)	18 (15.0)	<0.0001
Infestations	30 (41.1)	29 (24.1)	0.007
Contact dermatitis	1 (1.4)	3 (2.5)	0.619
Melasma	3 (4.1)	3 (2.5)	0.497
Infective dermatitis associated to HTLV-1	3	0	0.022
Other diagnoses	11 (15.0)	17 (14.1)	0.771
Normal findings on dermatological exam	9 (12.3)	37 (30.8)	0.005

HAM/TSP: HTLV-1-associated myelopathy/Tropical spastic paraparesis.

Infestations: Dermatophytosis (N: 55), Scabies (N;1), Pityriasis versicolor (N;3).

Other diseases: Herpes simplex (N: 4), Pytiriasis alba (N; 3), Cheilitis (N;3), Callosity (N: 2), Pressure ulcer (N:2), Verruca vulgaris (N:2); Vitiligo (N: 1), Photosensitivity (N: 1), Hypertrichosis (n: 1), Neurofibromatosis (N;1), Insect bite reaction (N;1), Acanthosis nigricans (N:1), Lichen planus (N:1), Nummular dermatitis (N:1), Basal cell carcinoma (N:1), Epidermal cyst (N:1), Rhinophyma (N:1), Miliaria (N:1), Rosacea (N:1), Exogenous ochronosis (N:1), Venous ulcer (N:1), Folliculitis (N:1), Lymphoma (N:1),

Hyperkeratosis plantare (N:1), Keratosis pilaris (N:1), Erythrodermia (N:1), Folliculitis decalvans (N:1), Pyogenic granuloma (N:1), Livedo reticularis (N:1), Madarosis (N:1), Erythema nodosum/Panniculitis (N:1), Molluscum contagiosum (N:1).

doi:10.1371/journal.pntd.0002546.t001

is necessary, leading in susceptible individuals to immune deregulation, with subsequent immunosuppression and superinfection with *Staphylococcus aureus* and beta-haemolytic streptococci, what additionally leads to chronic antigenic stimulation and persistent inflammation of the skin. Genetic, host and environmental factors have been shown to be associated [17].

Xerosis and acquired ichthyosis have been described as the main dermatological manifestations associated with HAM/TSP patients [11,12,18]. Xerosis is characterized by dryness of the skin, and acquired ichthyosis is clinically characterized by cutaneous xerosis and the formation of polygonal thin flat scales of varying sizes, mainly on the extremities [19].

Acquired ichthyosis is a consequence of hypohydrosis that may be secondary to the involvement of the autonomic nervous system, affecting directly the HTLV-1-infected skin cells [12,20]. On the basis of histopathological and immunohistochemical analyses of skin fragments of acquired ichthyosis from HAM/TSP individuals, it was concluded that keratinocytes are activated, probably as a result of cytokines that are liberated from HTLV-1-infected lymphocytes. This activation leads to an interference in keratinocyte differentiation and migration, resulting in a defect in the processes of normal desquamation, accumulation and retention of corneocytes [18]. We noticed that more than 50% of the SD-HTLV-1 cases involved xerosis/acquired ichthyosis and decided to include them in the same group because the acquired ichthyosis showed a similar clinical dermatological pattern, comparable with higher degree xerosis, and so they were clinically difficult to differentiate [12].

HTLV-1 proviral load is a laboratorial risk marker for the development of HAM/TSP and other diseases related to HTLV-1 infection [21,22]. The initial HTLV-1 proviral load was higher in the group with SD-HTLV-1 (p = 0.009) as well as in the SD-HTLV-1 HAM/TSP-free group (p = 0.021), both of which were statistically significant. Although the proviral load may differ greatly among individuals, it is relatively stable during the course of the HTLV-1-related disease [23]. In HAM/TSP patients this

Table 2. Characteristics of patients according to the presence of dermatological diseases associated to HTLV-1 infection.

Variable	Skin disorder associated to HTLV-1 (N=116)	No skin disorder associated to HTLV-1 (N=77)	p value
Age (years, mean, SD)	51±12	47±12	0.014
Female (%)	76	66	0.190
HCV coinfected (%)	13.9	14.3	1.000
Log ₁₀ HTLV-1 proviral load (copies/10 ⁴ PBMCs, mean, SD)	2.30±0.87*	1.75±1.00**	0.009
T CD4 cells count (cells/mm ³ , median, interquartile range)	1033 (766–1298)	982 (817–1285)	0.91
T CD8 cells count (cells/mm ³ , median, interquartile range)	557 (386–756)	548 (441–726)	0.97
HAM/TSP (%)	75	25	0.002

HCV: Hepatitis C virus; SD: Standard deviation; Skin disorders associated to HTLV-1: xerosis/acquired ichthyosis and seborrheic dermatitis.

*N tested = 67 subjects,

**N. tested: 42 subjects.

doi:10.1371/journal.pntd.0002546.t002

Table 3. Presence of skin disorders, epidemiological and laboratorial characteristics of the HTLV-1-infected subjects.

	Skin disorder associated	No skin disorder associated	p value
Variable	to HTLV-1(N=61)	to HTLV-1 (N = 59)	
Age (years, mean, SD)	51±13	44±12	0.002
Female (%)	55	45	0.167
HCV coinfected (%)	10.0	11.9	0.777
Log ₁₀ HTLV-1 proviral load (copies/10 ⁴ PBMCs, mean, SD)	2.05±0.86*	1.41±0.88**	0.021
T CD4 cells count (cells/mm ³ , median, interquartile range)	1035 (771.5–1318.5)	934 (814–1246)	0.38
T CD8 cells count (cells/mm ³ , median, interquartile range)	541 (378.25–747.75)	544 (432–703)	0.82

HCV: Hepatitis C virus; SD: Standard deviation; Skin disorders associated to HTLV-1: xerosis/acquired ichthyosis and seborrheic dermatitis. *N: 31,

**N: 31.

doi:10.1371/journal.pntd.0002546.t003

finding suggests that proviral load reaches a stable level determined by the relationship between viral expression and the immune response against the virus [23].

As previously shown, HTLV-1 was identified by PCR on skin cells in addition to lymphocytes in HTLV-1-infected persons, regardless of their clinical status ¹⁰. Because of these findings, several authors believe that HTLV-1 can modify the function of infected cells, resulting in skin disorders that are caused directly by the presence of the virus in the infected cells [10,16,24]. Another possible mechanism of skin disorders among HTLV-1-infected individuals is the production of cytokines in HTLV-1-infected lymphocytes, promoting a functional disturbance on skin cells [18]. This lack of association may be explained by the presence of HTLV-1 in specific sites that occur during HAM/TSP and the proviral load in the cerebral spinal fluid (CSF) [25]. Lezin et al. reported that the proviral load quantified in CSF was able to distinguish clearly between healthy groups of HTLV-1 carriers and patients presenting HAM/TSP [25].

Finally, for the first time, a high prevalence of skin disorders (76%) independent of clinical status was disclosed among HTLV-1-infected individuals. These findings may have important implications in the clinical setting in places where this infection is endemic and dermatologists, infectious diseases specialists and general clinicians should be aware of skin presentations of the HTLV-1 infection. Moreover, the influences of demography and

References

- LaGrenade L, Hanchard B, Fletcher V, Cranston B, Blattner W (1990) Infective dermatitis of Jamaican children: a marker for HTLV-I infection. Lancet 336:1345–1347.
- Osame M, Usuku K, Izumo S, Ijichi N, Amitani H, Igata A, et al. (1986) HTLV-I associated myelopathy, a new clinical entity. Lancet 1:1031–1032.
- Poiesz BJ, Ruscetti FW, Gazdar AF, Bunn PA, Minna JD, Gallo RC (1980) Detection and isolation of type C retrovirus particles from fresh and cultured lymphocytes of a patient with cutaneous T-cell lymphoma. Proceedings of the National Academy of Sciences of the United States of America 77:7415– 7459.
- Kawai H, Inui T, Kashiwagi S, Tsuchihashi T, Masuda K, Kondo A, et al (1992) HTLV-I infection in patients with autoimmune thyroiditis (Hashimoto's thyroiditis). J Med Virol38:138–141.
- Mochizuki M, Ono A, Ikeda E, Hikita N, Watanabe T, Yamaguchi K, et al (1996) HTLV-I uveitis. J Acquir Immune Defic Syndr Hum Retrovirol.Suppl 1:13 S50–S56.
- Ijichi S, Nakagawa M, Umchara F, Higuchi I, Arimura K, Izumo S, et al (1996) HAM/TSP: recent perspectives in Japan. J Acquir Immune Defic Syndr Hum Retrovirol.;13 Suppl 1:S26–S32.
- Carneiro-Proietti AB, Ribas JG, Catalan-Soares BC, Martins ML, Brito-Melo GE, Martins-Filho OA, et al (2002) Infection and disease caused by the human T cell lymphotropic viruses type I and II in Brazil. Revista da Sociedade Brasileira de Medicina Tropical 35:499–508.

co-morbid conditions may be relevant, but have not been fully studied. Thus, data derived from a cohort of referred patients followed in a specialized HTLV clinic may not be a representative sample of the whole population of HTLV-1 patients and therefore unlikely to reflect the prevalence of skin conditions among all HTLV-1 patients.

Supporting Information

Checklist S1 STROBE checklist. (DOCX)

Acknowledgments

We thank the personnel at the outpatient service of the Institute of Infectious Diseases Emilio Ribas for their excellent assistance. We also thank all our patients and Dr. Luis Augusto Marcondes Fonseca for critical review of this manuscript.

Author Contributions

Conceived and designed the experiments: JC RO JAS. Performed the experiments: RO JC JS ACPO. Analyzed the data: JC RO JAS JS ACPO. Contributed reagents/materials/analysis tools: JC RO. Wrote the paper: JC RO.

- Catalan-Soares B, Carneiro-Proietti AB, Proietti FA (2005) Heterogeneous geographic distribution of human T-cell lymphotropic viruses I and II (HTLV-I/II): serological screening prevalence rates in blood donors from large urban areas in Brazil. Cadernos de saude publica/Ministerio da Saude, Fundacao Oswaldo Cruz, Escola Nacional de Saude Publica 21:926–31.
- Gessain A, Cassar O (2012) Epidemiological Aspects and World Distribution of HTLV-1 Infection. Front Microbiol 3:1–23.
- Goncalves DU, Guedes AC, Proietti AB, Martins ML, Proietti FA, Lambertucci JR (2003) Dermatologic lesions in asymptomatic blood donors seropositive for human T cell lymphotropic virus type-1. The American Journal of Tropical Medicine and Hygiene 68:562–565.
- Bittencourt AL, de Oliveira Mde F (2010) Cutaneous manifestations associated with HTLV-1 infection. International Journal of Dermatology. 49:1099–1110.
- Lenzi ME, Cuzzi-Maya T, Oliveira AL, Andrada-Serpa MJ, Araujo AQ (2003) Dermatological findings of human T lymphotropic virus type 1 (HTLV-I)associated myelopathy/tropical spastic paraparesis. Clin Infect Dis 36:507–513.
- WHO (1989) Report of the scientific group on HTLV-1 and associated diseases, Kagoshima, Japan, December 1988. Kagoshima, Japan: World Health Organization.
- Novoa P, Penalva de Oliveira AC, Posada Vergara MP, da Silva Duarte AJ, Casseb J (2007) Molecular characterization of human T-cell lymphotropic virus type 2 (HTLV-II) from people living in urban areas of Sao Paulo city: evidence of multiple subtypes circulation. J Med Virol 79:182–187.

- Montanheiro PA, Oliveira AC, Posada-Vergara MP, Milagres AC, Tauil C, Marchiori PE, et al. (2005) Human T-cell lymphotropic virus type I (HTLV-I) proviral DNA viral load among asymptomatic patients and patients with HTLV-I-associated myelopathy/tropical spastic paraparesis. Brazilian Journal of Medical and Biological Research 38:1643–1647.
- Nobre V, Guedes AC, Proietti FA, Stanciolli E, Martins ML, Serufo JC, et al. (2005) Dermatologic lesions in patients infected with the human T-cell lymphotropic virus type 1 (HTLV-1). Revista da Sociedade Brasileira de Medicina Tropical 38:43–52.
- McGill NK, Vyas J, Shimauchi T, Tokura Y, Piguet V (2012) HTLV-1associated infective dermatitis: updates on the pathogenesis. Exp Dermatol 21:815–821.
- Milagres SP, Sanches JA, Jr., Milagres AC, Valente NY (2003) Histopathological and immunohistochemical assessment of acquired ichthyosis in patients with human T-cell lymphotropic virus type I-associated myelopathy. The British Journal of Dermatology 149:776–781.
- Hashiguchi T OM, Arimura K (1989) Skin manifestations in HTLV-1associated myelopathy (HAM): xerosis and erythema. In: Román CG VJ, Osame M, editors. HTLV-1 and Nervous System. New York: Alan R. p. 443–8.
- Alamy AH, Menezes FB, Leite AC, Nascimento OM, Araujo AQ (2001) Dysautonomia in human T-cell lymphotrophic virus type I-associated myclopathy/tropical spastic paraparesis. Ann Neurol;50:681–685.

- 21. Furtado Mdos S, Andrade RG, Romanelli LC, Ribeiro MA, Ribas JG, Torres EB, et al. (2012) Monitoring the HTLV-1 proviral load in the peripheral blood of asymptomatic carriers and patients with HTLV-associated myelopathy/ tropical spastic paraparesis from a Brazilian cohort: ROC curve analysis to establish the threshold for risk disease. J Med Virol84:664–671.
- Primo J, Siqueira I, Nascimento MC, Oliveira MF, Farre L, Carvalho EM, et al. (2009) High HTLV-1 proviral load, a marker for HTLV-1 associated myclopathy/tropical spastic paraparesis, is also detected in patients with infective dermatitis associated with HTLV-1. Brazilian journal of medical and biological research 42:761–764.
- Olindo S, Lezin A, Cabre P, Merle H, Saint-Vil M, Edimonana Kaptue M, et al. (2005) HTLV-1 proviral load in peripheral blood mononuclear cells quantified in 100 HAM/TSP patients: a marker of disease progression. Journal of the Neurological Sciences 237:53–59.
- Setoyama M, Kerdel FA, Elgart G, Kanzaki T, Byrnes JJ (1998) Detection of HTLV-1 by polymerase chain reaction in situ hybridization in adult T-cell leukemia/lymphoma. Am J Pathol 152:683–689.
- Lezin A, Olindo S, Oliere S, Varrin-Doyer M, Marlin R, Cabre P, et al. (2005) Human T lymphotropic virus type I (HTLV-I) proviral load in cerebrospinal fluid: a new criterion for the diagnosis of HTLV-I-associated myelopathy/ tropical spastic paraparesis? J Infect Dis 191:1830–1834.